Author's response to reviews

Title: What counts as reliable evidence for public health policy: the case of circumcision for preventing HIV infection

Authors:

Reidar K Lie (rlie@cc.nih.gov)
Franklin G Miller (FMiller@cc.nih.gov)

Version: 4 Date: 24 March 2011

Author's response to reviews:

Changed title and abstract
Removed title from above main body of text
Removed all tracking from manuscript